Author’s response to reviews

Title: Vocal local versus pharmacological treatments for pain management in tubal ligation procedures in rural Kenya: a non-inferiority trial

Authors:

Sarah C Keogh (sarahckeogh@gmail.com)
Kenzo Fry (kenzo.fry@mariestopes.org)
Edwin Mbugua (edwin.mbugua@mariestopes.or.ke)
Mark Ayallo (mark.ayallo@mariestopes.or.ke)
Heidi Quinn (heidi.quinn@mariestopes.org)
George Otieno (gotiochiengs@yahoo.co.uk)
Thoai D Ngo (thoai.ngo@mariestopes.org)

Version: 4  Date: 29 January 2014

Author’s response to reviews: see over
Dear Editor,

Thank you for your two comments related to page 5, which we have addressed as requested. We have modified the wording of the sentence about answering questions, and we have deleted the sentence about informed consent that was repeated.

We are also uploading a CONSORT checklist with our revised manuscript.

The reviewer did not have any additional requests or comments after our previous revision. We thank him for his time and thoughtful comments.

Many thanks for your time and attention,

Sarah Keogh (on behalf of all co-authors)